Author's response to reviews

Title: Adjuvant chemotherapy with gemcitabine and cisplatin compared to observation after curative intent resection of cholangiocarcinoma and muscle invasive gallbladder carcinoma (ACTICCA-1 trial) - a randomized, multidisciplinary, multinational phase III trial

Authors:

Alexander Stein (a.stein@uke.de)
Dirk Arnold (arnold@tumorbio.uni-freiburg.de)
John Bridgewater (j.bridgewater@ucl.ac.uk)
David Goldstein (d.goldstein@unsw.edu.au)
Lars Henrik Jensen (lars.henrik.jensen@rsyd.dk)
Heinz-Josef Klümpen (h.klumpen@amc.uva.nl)
Ansgar W Lohse (a.lohse@uke.de)
Björn Nashan (nashan@uke.de)
John Primrose (j.n.primrose@soton.ac.uk)
Silke Schrum (s.schrum@ctc-north.com)
Jenny Shannon (jenny.shannon@swahs.health.nsw.gov.au)
Eik Vettorazzi (e.vettorazzi@uke.de)
Henning Wege (hwege@uke.de)

Version: 2
Date: 15 April 2015

Author's response to reviews:

An e-mail responding the editorial request has been sent to BMCSeriesEditorial@biomecentral.com. The e-mail also states the requested changes.

Based on the requested changes a new cover letter has been uploaded (as suggested).